Author's response to reviews

Title: Development of an E-Supported Illness Management and Recovery Programme for Consumers with Severe Mental Illness Using Intervention Mapping, and Design of an Early Cluster Randomized Controlled trial.

Authors:

Titus AA Beentjes (titus.beentjes@radboudumc.nl)
Betsie GI van Gaal (betsie.vangaal@radboudumc.nl)
Peter JJ Goossens (p.goossens@dimence.nl)
Lisette Schoonhoven (L.Schoonhoven@soton.ac.uk)

Version: 5  Date: 28 July 2015

Author's response to reviews: see over
Dear Alfredo Auro V. Evangelista,

Thank you for letting us resubmit the study protocol (MS: 1749974992166260).

We did make the following changes to the manuscript:

1. We added line numbering in the document.
2. We included the title page in the document.
3. We changed the title a little to be more clear about the content of the article.
   The new title is:
   Development of an E-Supported Illness Management and Recovery Programme for
   Consumers with Severe Mental Illness Using Intervention Mapping, and Design of an Early
   Cluster Randomized Controlled trial.
4. We changed the names of the sections in the abstract into:
   Background, Method, Design, and Discussion.
5. We changed the names of the sections in the article into:
   Background, Method, Design, and discussion
   In the Method section we included the results of the development steps 1 to 5.
   The Design section is the result of step 6 of the development process.
6. On page 16 we added a the issue Competing interests, we declare we have no competing
   interests.
7. We added a description of the authors’ contribution on page 16.
8. We added a description of acknowledgements on page 17

We hope you will allow the article for editorial assessment.

Kind regards,
Titus Beentjes
Junior-researcher
Scientific Institute for Quality of Healthcare (IQ healthcare)
titus.beentjes@radboudumc.nl
M +31(0)6 51284459
Radboud universitair medisch centrum
PO Box 9101, 114 IQ healthcare
6500 HB Nijmegen